Author’s response to reviews

Title: Clinical Placement Experiences by Undergraduate Nursing Students in Selected Teaching Hospitals in Ghana.

Authors:

CONFIDENCE ATAKRO (confidenceatakro@gmail.com;caatakro@csuc.edu.gh)
Ernestina Armah (earmah@csuc.edu.gh)
Awube Menlah (amenlah@yahoo.com)
Isabella Darko (darkoisabella@gmail.com)
Stella Stella (sbaddo@csuc.edu.gh)
Peter Adatara (awakape@yahoo.com)
George Boni (sedi.boni24@gmail.com)

Version: 1 Date: 30 Jul 2018

Author’s response to reviews:

Editor comment 1: Please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate in Ethics approval and consent to participate.

Response 1: Full name of ethics committee and institution provided as directed.

Editor comment 2: Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

Response 2: Email addresses of all authors now provided on title page. Corresponding author indicated.

Editor comment 3: Please provide an Authors' Contributions section in Declarations.

Response 3: Authors' Contributions section now provided under declaration.
Editor comment 4: Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Response 4: This has been complied with.

Editor comment 5: Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Response 5: This has been done and ensured.

Editor comment 6: Please include an Abbreviations section after Conclusions.

Response 6: This has now been done.

Editor comment 7: The Availability of data and materials section refers to the raw data used in your study and presenting tables and figures is not sufficient to state that all data is contained within the manuscript and additional files. Please only use this statement if you have indeed provided all raw data on which your study is based. We strongly encourage all authors to share their raw data, either by providing it in a supplementary file or depositing it in a public repository and providing the details on how to access it in this section. If you do not wish to share your data, please clearly state this in this section along with a justification. Data availability statements can take one of the following forms (or a combination of more than one if required for multiple datasets):

- The datasets generated and/or analysed during the current study are available in the [NAME] repository, [PERSISTENT WEB LINK TO DATASETS]
- The datasets used and/or analysed during the current study available from the corresponding author on reasonable request.
- The datasets generated and/or analysed during the current study are not publicly available due [REASON WHY DATA ARE NOT PUBLIC] but are available from the corresponding author on reasonable request.
- Data sharing is not applicable to this article as no datasets were generated or analysed during the current study.
- The data that support the findings of this study are available from [third party name] but restrictions apply to the availability of these data, which were used under license for the current study, and so are not publicly available. Data are however available from the authors upon reasonable request and with permission of [third party name].

Response 7: This has been complied with.
Editor comment 8: Please remove the cover letter from the file inventory.

Response 8: This has been removed as directed.

Editor comment 9: Please, provide the corresponding author details in the title page, not as a supplementary file.

Response 9: This has been complied with.

Editor comment 10: Please read the following information and revise your manuscript as necessary. If your manuscript does not adhere to our editorial requirements this will cause a delay whilst the issue is addressed. Failure to adhere to our policies may result in rejection of your manuscript.

In accordance with BioMed Central editorial policies and formatting guidelines, all submissions to BMC Nursing must have a Declarations section which includes the mandatory sub-sections listed below. Please refer to the journal's Submission Guidelines web page for information regarding the criteria for each sub-section (https://bmcnurs.biomedcentral.com/). Where a mandatory section is not relevant to your study design or article type, for example, if your manuscript does not contain any individual persons data, please write "Not applicable" in these sections.

For the 'Availability of data and materials' section, please provide information about where the data supporting your findings can be found. We encourage authors to deposit their datasets in publicly available repositories (where available and appropriate), or to be presented within the manuscript and/or additional supporting files. Please note that identifying/confidential patient data should not be shared. Authors who do not wish to share their data must state that data will not be shared, and provide reasons for this in the manuscript text. For further guidance on how to format this section, please refer to BioMed Central's editorial policies page - http://www.biomedcentral.com/submissions/editorial-policies#availability+of+data+and+materials.

Declarations

- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors' Contributions
- Acknowledgements
- Authors' Information

Response 10: This has been complied with.